Multilocus Genotyping of Human *Giardia* Isolates Suggests Limited Zoonotic Transmission and Association between Assemblage B and Flatulence in Children

Marianne Lebbad^{1,2}, Ingvor Petersson³, Lillemor Karlsson⁴, Silvia Botero-Kleiven¹, Jan O. Andersson⁵, Bo Svenungsson³, Staffan G. Svärd^{2,6}*

1 Department of Diagnostics and Vaccinology, Swedish Institute for Communicable Disease Control, Solna, Sweden, 2 Microbiology and Tumor Biology Centre, Karolinska Institutet, Stockholm, Sweden, 3 Department of Communicable Disease Control and Prevention, Stockholm County Council, Stockholm, Sweden, 4 Department of Clinical Microbiology, Karolinska University Hospital, Stockholm, Sweden, 5 Department of Molecular Evolution, Evolutionary Biology Centre, Uppsala University, Uppsala, Sweden, 6 Department of Cell and Molecular Biology, Uppsala University, Uppsala, Sweden

Abstract

Background: Giardia intestinalis is one of the most common diarrhea-related parasites in humans, where infection ranges from asymptomatic to acute or chronic disease. *G. intestinalis* consists of eight genetically distinct genotypes or assemblages, designated A–H, and assemblages A and B can infect humans. Giardiasis has been classified as a possible zoonotic disease but the role of animals in human disease transmission still needs to be proven. We tried to link different assemblages and sub-assemblages of *G. intestinalis* isolates from Swedish human patients to clinical symptoms and zoonotic transmission.

Methodology/Principal Findings: Multilocus sequence-based genotyping of 207 human *Giardia* isolates using three gene loci: ß-giardin, glutamate dehydrogenase (*gdh*), and triose phosphate isomerase (*tpi*) was combined with assemblage-specific *tpi* PCRs. This analysis identified 73 patients infected with assemblage A, 128 with assemblage B, and six with mixed assemblages A+B. Multilocus genotypes (MLGs) were easily determined for the assemblage A isolates, and most patients with this genotype had apparently been infected through anthroponotic transmission. However, we also found evidence of limited zoonotic transmission of *Giardia* in Sweden, since a few domestic human infections involved the same assemblage A and it was also more common in patients with suspected treatment failure. However, a large genetic variability made determination of assemblage B MLGs problematic. Correlation between symptoms and assemblages was found only for flatulence, which was significantly more common in children less than six years of age infected with assemblage B.

Conclusions/Significance: This study shows that certain assemblage A subtypes are potentially zoonotic and that flatulence is connected to assemblage B infections in young children. Determination of MLGs from assemblages A and B can be a valuable tool in outbreak situations and to help identify possible zoonotic transmission.

Citation: Lebbad M, Petersson I, Karlsson L, Botero-Kleiven S, Andersson JO, et al. (2011) Multilocus Genotyping of Human Giardia Isolates Suggests Limited Zoonotic Transmission and Association between Assemblage B and Flatulence in Children. PLoS Negl Trop Dis 5(8): e1262. doi:10.1371/journal.pntd.0001262

Editor: Daniel Eichinger, New York University School of Medicine, United States of America

Received January 12, 2011; Accepted June 21, 2011; Published August 2, 2011

Copyright: © 2011 Lebbad et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This study was supported by FORMAS, VR-M, SLL and SMI. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: Staffan.svard@icm.uu.se

Introduction

Giardia intestinalis (synonyms: *G. lamblia*, *G. duodenalis*) is a protozoan parasite that infects a wide array of vertebrates, including humans, pets, livestock, wildlife, and marine animals [1,2]. *Giardia* has a global distribution and is one of the most common diarrhea-related parasites in humans, where infection ranges from asymptomatic to symptomatic, involving both acute and chronic disease. According to estimates, about 200 million people worldwide have symptoms of intestinal giardiasis, and 500 000 new cases occur annually [3]. Due to its impact on health, especially among children in developing countries, *Giardia* has been included in the Neglected Diseases Initiative of the World Health Organization (WHO) since 2004 [4].

Giardia intestinalis consists of eight morphologically identical but genetically distinct genotypes or assemblages, designated A–H [1,5]. Assemblages A and B can infect humans and other mammals, whereas assemblages C–H appear to be host specific. Giardiasis has been classified as a possible zoonotic infection by the WHO since 1979 [6], and studies conducted in India and Thailand [7,8] have suggested zoonotic transmission of *Giardia*; nonetheless, the role of animals in human disease transmission still needs to be proven [9–11]. Several investigations have also tried to link the severity of infection to a certain assemblage, but the results have been inconclusive [12–16]. Most of those studies relied on genotyping using only one or two genetic markers, such as the small subunit ribosomal RNA (*ssrRNA*), β-giardin, glutamate

Author Summary

Giardia intestinalis is a protozoan parasite found worldwide and it is a major cause of diarrhea in humans and other mammals. The genetic variability within *G. intestinalis* is high with eight distinct genotypes or assemblages (A-H). Here we performed sequence-based multilocus genotyping of around 200 human *Giardia* isolates. We found evidence of limited zoonotic transmission of certain A subtypes and an association between flatulence and assemblage B infection in children. This shows that it is important to investigate different assemblages and subassemblages of *G. intestinalis* in human infections in order to understand the clinical significance, zoonotic potential, sequence divergence, and transmission pathways of this parasite.

dehydrogenase (gdh), or triose phosphate isomerase (tpi) gene. The information thus obtained has limited discriminatory power, and it has recently been suggested that a special multilocus genotyping approach should be used to compare multilocus genotypes (MLGs) from different sources [17]. Lately, the occurrence of mixed assemblage infections has also gained attention through research in which use of assemblage-specific tpi primers has allowed much more extensive detection of mixed assemblage A and B infections than is feasible when using more general primers [18,19]. Accordingly, in-depth studies that focus on several genetic loci and use assemblage-specific primers are needed to clarify both the issue of zoonotic transmission and the correlation between assemblage and disease pattern. It might also be possible to use this combination of methods for source tracing in outbreak situations.

In Sweden, giardiasis has been a notifiable disease since 1989, and 1200–1500 cases are reported each year in a population of about nine million. Although most cases are imported, domestic transmission occurs as well. Recently, a study of *Giardia* genotypes in animals in Sweden was published [20], but molecular characterization of human *Giardia* isolates in this country has been limited to an outbreak involving three nursery schools, where the only subtype found was A3, as determined at the β -giardin locus [21].

The purpose of the present study was to determine the genetic variability of *G. intestinalis* isolated from patients with infections acquired in Sweden and abroad. Multilocus genotyping was used as a tool to investigate *Giardia* with regard to the relationship between assemblages and symptoms, the zoonotic potential, sequence divergence, and possible transmission dynamics.

Materials and Methods

Sources of isolates and patient information

Fecal samples from 214 patients with *Giardia* infection diagnosed by light microscopy at Karolinska University Hospital, Stockholm, between May 2007 and April 2009 were referred to the Swedish Institute for Communicable Disease Control (SMI) in Solna. The majority of the study participants (including nine adopted children) had consulted a physician due to intestinal symptoms (n = 175). The remaining cases (n = 39) were detected in connection with health check-ups (adopted children n = 9, other subjects n = 10) or through source tracing (n = 20). In addition, 11 patients provided a second fecal sample that was microscopically positive after treatment. Besides routine parasitological examination of all samples, 179 of the fecal specimens were also cultured for bacterial enteropathogens by standard methods. Information regarding travel abroad within two weeks prior to onset of disease, symptoms, treatment, and possible routes of transmission was obtained by use of postal questionnaires or telephone interviews.

Ethics statement

The study was approved by the Regional Ethics Committee of Karolinska Institutet, Stockholm, Sweden. Written consent for being part of the study was obtained from each patient and parents or primary caretakers signed for their children.

Microscopy and DNA extraction

The *Giardia* cysts were studied after concentration by the formol/ethyl acetate technique, using direct immunofluorescent labeling with a monoclonal antibody (Agua-Glo, Waterborne Inc., New Orleans, LA, USA) and counterstaining with DAPI (4'6-diamidino-2-phenyl-indole). DNA was extracted directly from stools using a QIAamp DNA Mini Kit (Qiagen, Hilden, Germany) according to the manufacturer's recommendations. Before carrying out the extraction procedure, the cysts were disrupted in a Mini-BeadBeater (Biospec Products Inc., Bartlesville, OK, USA) [22]. For a limited number of samples (n = 13) showing no initial amplification, the cysts were isolated on a sucrose gradient and subjected to a renewed extraction [22].

Molecular methods

All 225 samples were analyzed using a nested ß-giardin PCR [23] a semi-nested gdh PCR [24], and a nested thi PCR [25], with the expected amplicons of 511, 432, and 530 bp, respectively. Using conditions described elsewhere, restriction fragment length polymorphism (RFLP) analysis was performed on aliquots of the products from all three PCR amplifications: B-giardin [23] gdh [24], and thi [26]. All samples were also analyzed using assemblage A- and B-specific tpi PCR primers [19,27] with the expected amplicons of 373 and 400 bp, respectively. All amplicons generated in the ß-giardin PCR and the majority of amplicons from the gdh and thi PCRs were sequenced in both directions. In addition, an alternative nested gdh PCR with subsequent sequencing as well as sequencing of the amplicons obtained from single ß-giardin PCR was performed on a limited number of assemblage A isolates [17,28]. The sequences thus obtained were used to create a concatenated assemblage A tree. Chromatograms and sequences were examined using the BioEdit sequence analysis program (http://www.mbio.ncsu.edu/BioEdit/page2. html). The BLAST tool (http://www.ncbi.nlm.nih.gov/blast/) was used to compare nucleotide sequences with sequences in the GenBank database.

Representative nucleotide sequences without ambiguous positions have been deposited in GenBank under the following accession numbers: GQ329671-GQ329679, HM1407-HM140725, HM136880-HM136891, HM165208-HM165227, and JF773747-JF773759. In addition all β-giardin, *gdh*, and *tpi* sequences from 120 assemblage B isolates, positive at all three loci, have been appended as Supplementary Files S1, S2, S3.

Phylogenetic analysis

For phylogenetic analyses, MLGs with unambiguous sequences identified in the current investigation and in our previous study of animals in Sweden [20] were combined with reference sequences of representative isolates from GenBank (Supplementary Table S1). After removal of primer sequences, the ß-giardin, *gdh*, and *tpi* datasets for assemblage A consisted of 700 (product from single ß-giardin PCR), 694 (merged products from two *gdh* PCRs), and 490 nucleotides, respectively, and the corresponding datasets for

assemblage B comprised 475, 393, and 490 nucleotides. RAxML version 7.0.4 [29] was used to perform maximum likelihood analyses with the GTR substitution model and among-site rate variation (GTRGAMMA), and the same approach was applied in bootstrap analyses with 1000 replicates.

Statistical methods

Chi squared test or Fisher's exact test, as appropriate, were used to evaluate differences between characteristics of patients infected with different assemblages.

Results

Combined results of ß-giardin, gdh, and tpi PCR amplifications

Altogether, *Giardia* isolates from 207 of the 214 patients were successfully genotyped. PCR analyses yielded the expected amplicons for all three genes in 192 isolates, for two genes in nine isolates, and for one gene in six isolates. Seven isolates were negative in all three PCRs despite repeated trials. Six of these seven exhibited only a few cysts, which were all DAPI negative; the seventh sample did contain DAPI-positive cysts, but it had most likely been exposed to formalin before extraction. Of the 11 additional samples that were microscopically positive after treatment, 10 were positive for all three genes, whereas one sample was negative.

PCR-RFLP and assemblage A- and B-specific tpi PCR

RFLP analysis was successful for all isolates amplified at the ßgiardin, *gdh*, and *tpi* loci (Table 1). Most of the isolates showed the expected RFLP patterns for assemblages A and B at all three genes. Patterns corresponding to mixed assemblage A and B infections were observed in three isolates (Table 2). Three other isolates (Sweh166, 173, and 178) that had a sub-assemblage AI pattern in the *gdh* RFLP displayed a novel assemblage A pattern at the ß-giardin gene (Fig. 1). This pattern has recently been observed in samples from ruminants and cats in Sweden (unpublished data). Considering assemblage B, one new pattern was observed at the ßgiardin gene (Sweh198), and two new patterns were found at the *tpi* gene (Sweh121 and Sweh154) (Fig. 1). Additional bands in the RFLPs of assemblage B were seen in all three genes: ß-giardin (n = 1), *gdh* (n = 11), and *tpi* (n = 9) (Fig. 1). Sequence analyses of

 Table 1. Distribution of assemblages among 207 isolates

 determined by PCR-RFLP and assemblage A- and B-specific

 PCR.

	Assei	mblages			
	A	в	A+B	Not amplified	Total positive
ß-giardin RFLP	74 ^a	124 ^b	2	14	200
gdh RFLP	73 ^c	129	0	12	202
tpi RFLP	72	124 ^d	2	16	198
tpi A+B PCR	70	122	6	16	198
Combined results	73	128	6	7	207

^alncluding three samples with a novel A pattern (Sweh166, 173, and 178). ^blncluding one sample with a novel B pattern (Sweh198).

Concluding three samples with AI (Sweh166, 173, and 178) and 70 with AII patterns.

^dIncluding two samples with novel B patterns (Sweh121 and 154). doi:10.1371/journal.pntd.0001262.t001 these isolates revealed nucleotide substitutions or overlapping nucleotides (double peaks in chromatograms) at different restriction sites, which explains both the new patterns and the additional bands. The results obtained using the assemblage A- and B-specific tpi primers confirmed all findings of the RFLP analysis of the tpigene and added three more mixed A+B infections that were not detected in any of the RFLPs (Tables 1 and 2). The additional findings were confirmed by sequencing the amplicons from the assemblage-specific tpi PCR. All 16 samples that were negative with the original tpi primers were also negative with the assemblage-specific tpi PCR demonstrated 73 patients with assemblage A, 128 with assemblage B, and six with mixed assemblages A+B (Tables 1 and 2).

Sequencing and phylogenetic analyses

In total, 200 β -giardin, 195 gdh, and 195 tpi amplicons were sequenced, including all the 192 isolates that were positive at all three genes.

Assemblage A

Sequencing of 74 assemblage A isolates at the B-giardin locus revealed 36 isolates with subtype A2, 30 with subtype A3, and five with a mixture of subtype A2 and A3, as demonstrated by overlapping nucleotides at subtype specific positions in the chromatograms. Furthermore, three assemblage A isolates that exhibited a new ß-giardin RFLP assemblage A pattern had sequences identical to GenBank acc. no. AY655702, previously reported from cats, ferrets and various ruminants [17,20,30-32]. At the tpi locus, these three isolates corresponded to GenBank acc. no. EU781027, a sequence identified in different animal species in Sweden [20]. However, at the gdh locus, the three sequences were either identical to sub-assemblage AI or to GenBank acc. no. EU769224, which was recently found in a fallow deer and a cat in Sweden [20]. In addition, one assemblage A isolate at the tpi locus had a sequence that corresponded to the isolate ISSGd85 (GenBank acc. no. EU041753) obtained from a human in Western Sahara [26]. All remaining assemblage A isolates that were sequenced at the gdh (n = 69) and tpi (n = 67) loci corresponded to sub-assemblage AII. Apart from the five isolates with mixed A2-A3 subtypes at the ß-giardin gene and one isolate with overlapping nucleotides at position 445 of the *tpi* gene, all other assemblage A isolates had sequences without ambiguous positions (Table 3).

Multilocus genotyping of assemblage A isolates

Five different assemblage A MLGs could be identified in 67 isolates that had single assemblage A infections and had been sequenced on all three genes (Table 3). Two of these, MLG AII-1 and MLG AII-2, were found most frequently, whereas the other three were seen in only a few cases. A phylogenetic analysis was performed on these five MLGs, along with two reference MLGs (AI and AIII) and three MLGs unique to our previously conducted genotyping of animals in Sweden [20] (Table 3 and Fig. 2A). Three of the MLGs (from Swch166, 178, and 173), which are shared with animals, could not be conclusively categorized as AI or AII (Fig. 2A).

Assemblage B

A much more complex picture was found for assemblage B than for assemblage A. At the ß-giardin locus, 59 of 124 (48%) sequenced assemblage B isolates exhibited no double peaks at any position in the chromatograms, and subtypes could be determined. Table 2. Molecular characterization of isolates from patients presenting with mixed assemblage A and B infections.

Isolate	ß-giardin ^a	tpi ^a	gdhª	tpi A+B PCR	Origin of infection	Co-infections	Patient information
Sweh068	B ^b	B ^b	B ^b	A ^c +B	Ethiopia	Campylobacter	Adoptive child
Sweh098	A3	All	All	A+B ^c	India	Blastocystis	Adult tourist
Sweh110	A+B	A+B	B ^b	A ^c +B	India	Blastocystis	Adult tourist
Sweh131	A2	A+B	All	A+B ^c	China	None	Adoptive child
Sweh140	A3	All	All	A+B ^c	India	Shigella and Campylobacter	Adult tourist
Sweh207	A+B	Bb	Bb	A ^c +B	India	Blastocystis	Adult tourist

^aData based on RFLP and sequencing.

^bSequences containing overlapping nucleotides.

^cConfirmed by sequencing.

doi:10.1371/journal.pntd.0001262.t002

These 59 isolates corresponded to 10 previously reported and 10 new subtypes. The novel subtypes were represented by one isolate each, whereas the two previously reported subtypes B1-1 and B1-3 (GenBank acc. nos. EU637579 and EU881698; subtype names as suggested by Geurden and collegues [18]) were found in nine and 18 patients, respectively. At the tpi locus, 50 out of 122 (41%) sequenced assemblage B isolates had no double peaks in the chromatograms. Sixteen different subtypes were identified, seven of which had already been described and nine were novel. The most common subtype was found in 21 isolates and corresponded to BIV isolate Ad-19 (GenBank acc. no. AF069560). Of the 123 assemblage B isolates that were sequenced at the gdh locus, 44 (36%) displayed no double peaks in the chromatograms, whereas the remaining 79 exhibited from one to 14 positions with overlapping nucleotides. Twelve different subtypes were seen, five of them previously reported and seven new. Two subtypes were frequently seen and corresponded to two different BIV isolates, in 18 cases Ad-7 (GenBank acc. no. L40508) and in 12 cases gdber12 (GenBank acc. no. DQ923581). The remaining 10 subtypes were seen in one to four isolates each.

Multilocus genotyping of Assemblage B isolates

Thirty-one of 120 assemblage B isolates that were successfully sequenced at all three loci and without overlapping nucleotides at any position were grouped into 14 different MLGs (Table 4). Phylogenetic analyses of individual genes (Supplementary Fig. S1) and concatenation of the three genes (Fig. 2B) indicated that MLGs 12, 13, 14 and the monkey isolate (Monkey200) branched together to the exclusion of the other human MLGs, the guinea pig (Guinea pig138) and the rabbit (Rabbit176) MLGs, with 100% bootstrap support (Fig. 2B). In the phylogenies of *gdh* and *tpi*, MLGs 12–14 and Monkey200 branched together with BIII isolates, whereas the other MLGs branched with BIV isolates, Guinea pig138 and Rabbit176 (Supplementary Fig. S1B and C).



β-giardin

tpi

gdh

Figure 1. PCR-RFLP analysis of Giardia isolates. Panel A: Gelred (Biotium) stained 3.5% MetaPhor agarose gel (Cambrex) showing electrophoretic separation of nested β -giardin PCR products (511 bp) after digestion with HaellI: lane 1, assemblage A (novel pattern Sweh166, 173, 178); lane 2, assemblage A (ordinary pattern); lane 3, assemblage B (ordinary pattern), lane 4, assemblage B (novel pattern Sweh198); lane 5, assemblage B (mixed pattern). Panel B: Electrophoretic separation of nested *tpi* PCR products (530 bp) after digestion with Ddel: lane 1, assemblage A; lane 2, assemblage B (ordinary pattern); lane 3, assemblage B (novel pattern Sweh121), lane 4, assemblage B (novel pattern Sweh154); lanes 5, 6 and 7, assemblage B (mixed patterns). Panel C: Electrophoretic separation of semi-nested *gdh* PCR products (432 bp) after digestion with NlaIV: lane 1, assemblage All; lane 2, assemblage Al; lanes 3 and 4, assemblage B; lanes 5 and 6, assemblage B (mixed patterns). Molecular size markers (M) are 50-bp ladders (Invitrogen).

doi:10.1371/journal.pntd.0001262.g001

Table 3. Characterization of 67 assemblage A isolates* based on sequencing data and assemblage-specific PCR.

No. of isolates (isolate code)	ß-giardin	tpi	gdh	tpi A+B	MLG
31	A2	All	All	А	All-1
1 (Sweh037)	A2	All ^a	All	А	Mixed
1 (Sweh038)	A2	All ^b	All	А	All novel
26	A3	All	All	А	All-2
5	A2+A3 ^c	All	All	A	Mixed All-1+ All-2
2 (Sweh166, 178)	A ^d	Ae	AI	А	A novel ^g
1 (Sweh173)	A ^d	Ae	A ^f	А	A novel ^h

*Isolates with mixed assemblage A+B infections are not included.

^aA/G at position 445.

^bGenBank acc. no. EU041753 (isolate ISSGd85, human).

^cC/T at position 460 and 468.

^dGenBank acc. no. AY655702 (unnamed isolate, cattle).

^eGenBank acc. no. EU781027 (isolate Swecat170, cat). ^fGenBank acc. no. EU769224 (isolate Swecat202, cat).

^gEqual to previous MLG from various Swedish animals (cats and ruminants).

^hEqual to previous MLG from Swefd154 (fallow deer) and Swecat202 (cat).

doi:10.1371/journal.pntd.0001262.t003

The separation into BIII and BIV was not clear in the ß-giardin phylogeny (Supplementary Fig. S1A), although the bootstrap support for the alternative topology was low. Table 4 shows that a few "subtypes" from each locus (B1-1, B1-3, and B1-5 at Bgiardin; Ad-19 and GS/M at tpi; Ad-7, gd-ber 12, and Vanc89/ UBC/059 at gdh) were identified in many of the MLGs, but the grouping was not consistent, because they appeared in different combinations.

Allelic sequence divergence in Assemblage B

Cat035

I Sheep015

AI (Sheep055)

95

Cat078

Overlapping nucleotides were frequently observed in assemblage B sequences and occurred at the ß-giardin gene in 52% of the isolates, at the *tpi* gene in 59%, and at the *gdh* gene in 64%. In Supplementary Tables S2A-C, the present B-giardin, tpi, and gdh

Sweh040

L Sweh173; Fallow deer 165

99 All-1 (31 isolates)

_ All-2 (26 isolates)

0.01

sequencing data from 120 assemblage B isolates are compared with corresponding data from reference BIII and BIV isolates. Double peaks occurred at 47 different positions in the sequenced 511-bp fragment of the ß-giardin gene, at 88 positions in the 530bp fragment of the tpi gene, and at 59 positions in the 432-bp fragment of the gdh gene; note that only those positions where nucleotide substitutions or overlapping nucleotides occurred in five or more isolates are shown in the tables. Supplementary Tables S2A-C illustrate widespread occurrence of overlapping nucleotides in the positions that Wielinga and Thompson [33] have proposed to differentiate between sub-assemblage BIII and BIV. Supplementary Table S2A shows that 30% of the isolates (36/120) had C and T at position 354 of the B-giardin gene, the only position suggested to differentiate between BIII and BIV in this locus [19]. Considering the tpi gene, the five positions 39, 91, 165, 168, and 210 have been proposed for differentiation [33], and in our study double peaks were frequently observed for all of these positions in the chromatograms (24%, 27%, 33%, 24%, and 12%, respectively; Supplemenatry Table S2B). At the gdh gene positions 309, 429, 447, 540, 561 and 612 have been suggested for subassemblage differentiation [33], but again we found overlapping nucleotides in all those positions (38%, 32%, 36%, 46%, 18%, and 41%, respectively; Supplementary Table S2C).

Patient data

The ages of the patients ranged from 0 to 76 years (median 31.5, mean 30.4); 109 were female and 105 male. The distribution of assemblages in relation to gender among the 207 patients whose isolates were successfully genotyped was as follows: 36 females had assemblage A, 64 assemblage B, and four mixed assemblages A+B; among the males, 37 had assemblage A, 64 assemblage B, and two mixed assemblages A+B. The age distribution was as follows: 0-5 years (n = 40), 6–10 years (n = 19), 11–20 years (n = 9), and >20 years (n = 146). Clinical and epidemiological data were obtained from 181 patients in the form of survey responses. For the remaining 33 participants, designation of the country of origin of the infection and partial information concerning symptoms were acquired from the mandatory notification submitted by the clinician according to the Swedish Communicable Diseases Act.



Figure 2. Nucleotide maximum likelihood trees based on concatenated datasets for ß-giardin, gdh, and tpi gene sequences. MLGs with unambiguous sequences identified in this study (Table 3 and 4) are combined with reference isolates and isolates from our previous MLG study of animals in Sweden [20] (Supplementary Table S1). (A) Phylogenetic tree based on 1884 aligned positions of assemblage A isolates. Isolates identified in the present study are indicated in red. (B) Phylogenetic tree based on 1358 aligned positions of 17 assemblage B MLGs identified in our present and our previous study [20]. BIII (red) and BIV (blue) isolates are assigned according to their clustering with reference isolates in phylogenetic trees of the individual genes (Fig. S1). Only bootstrap support values >50% are shown. doi:10.1371/journal.pntd.0001262.g002

Isolate	lsolate/Subtype/GenBa	ink accession number		DIM	Intestinal symptoms	Origin of infection
	ß-giardin	tpi	gdh			
Sweh001 Sweh021, 022 ^a Sweh021, 056, 057, 058 ^a Sweh156 Sweh159, 160 ^a Sweh213 Sweh217	B1-3 EU881698	Ad-19/BIV AF069560	Ad-7/BIV L40508	-	No Yes, Yes Yes, No, No, No Yes, No Yes Yes	Switzerland Sweden, Malta ^b Sweden Spain Sweden Sweden USA
Sweh059	B1-3	Ad-19/BIV	gd-ber12, DQ923581	2	Yes	Finland
Sweh144	B1-3	GS/M/BIV/L02116	gd-ber12	ñ	Yes	Sweden
Sweh158 Sweh199, 200, 202 ^ª	B1-1 EU637579	Novel B HM140723	Ad-7/BIV	4	Yes Yes, No, No	USA Sweden
Sweh168	B1-1	Ad-19/BIV	gd-ber12	5	Yes	Sweden
Sweh154	B1-1	M12/BIV EU834845	gd-ber12	6	Yes	Sweden
Sweh074	B1-1	Ad-19/BIV	Ad-7/BIV	7	Yes	Mauritius
Sweh179	B1-1	GS/M BIV	Vanc89/UBC/059 AY178750	8	Yes	Nicaragua
Sweh033	B1-2 EU881697	ST25 DQ789114	Vanc89/UBC/059 BIV	6	Yes	Canada
Sweh192	B1-5 EU881700	Ad-19/BIV	gd-ber12	10	Yes	Sweden
Sweh047, 048, 049 ^a	B1-5	GS/M/BIV	gd-ber12	11	Yes, Yes, No	Sweden
Sweh060	BG-Ber6 DQ090527	2434 AY368165	Novel B HM136881	12	Yes	Sweden
Sweh136	ISSGF4/B4 AY072728	BAH12/BIII AF069561	Novel B HM136884	13	Yes	Germany
Sweh107	Novel B HM165222	Novel B HM140716	BAH12/BIII AF069059	14	Yes	India
*The sequences contained no ambiguous po:	ssitions at any loci. Identical	sequences are marked in the same	color.			

Table 4. Characterization of 31 assemblage B isolates based on the B-giardin, *gdh*, and *tpi* gene sequences^{*}.

^aFamily cluster. ^bCountry of infection for index case. doi:10.1371/journal.pntd.0001262.t004

Origin of infection

Table 5 shows the distribution of assemblages in relation to country of infection. Fifty-three patients with symptoms of giardiasis did not report any travel outside Sweden within 14 days prior to the onset of their symptoms and were thus considered to represent domestic infections, and 13 asymptomatic patients were also regarded as domestic infections based on previous history of travel. The majority of the patients (61%) were infected outside Europe and more than half of those in various Asian countries. Assemblage B was the most prevalent assemblage from all parts of the world except Latin America, for which there was more equal distribution of assemblages A and B (Table 5).

Microbiological investigation

Bacterial investigations performed on 179 samples revealed that 12 patients had co-infections with the following bacteria: *Campylobacter* (n = 7), *Shigella* (n = 2), *Salmonella* (n = 1), and both *Campylobacter* and *Shigella* (n = 2). Parasitological examination of all samples detected several additional parasites: *Enterobius vernicularis* (n = 3), *Trichuris trichiura* (n = 1), *Hymenolepis nana* (n = 1), *Cryptosporidium* spp. (n = 4), *Blastocystis* spp. (n = 31), *Entamoeba dispar* (determined by PCR) (n = 2), and other non-pathogenic amoebas (n = 26).

Correlation between assemblages and symptoms

Table 6 presents clinical data from 145 symptomatic persons who responded to the survey and had no co-infections with other diarrhea-related enteropathogens. The only correlation found between symptoms and infection with specific assemblages was noted for flatulence, which was reported more frequently by patients infected with assemblage B (p = 0.006, Table 6). However, this correlation was no longer apparent when the analysis was restricted to patients over five years of age (p = 0.16, data not shown). Separate analyses of symptomatic children aged 0-5 years (n = 12) revealed six children that were infected with assemblage B, and they all suffered from flatulence; six had assemblage A infection, and none of them experienced flatulence. Thus, flatulence was significantly more common in children infected with assemblage B (p = 0.0022). Ten patients, two of whom were co-infected with Campylobacter and Shigella, were hospitalized due to diarrhea and treated with intravenous fluids. Seven of these subjects were infected with assemblage B and two with assemblage A, and one was infected with assemblages A+B. The two patients with assemblage A (Sweh166 and 178) were both infected in Sweden with a unique MLG hitherto found only in animals in this country (Table 3).

Family clusters

Eighteen family clusters involving a total of 46 individuals (2-5 per cluster) were identified. In 16 of these clusters, samples from

two or more individuals were successfully genotyped. Six families were infected with assemblage A and eight with assemblage B, whereas in two families the members were infected with either assemblage A or B. Multilocus genotyping was successful in all six assemblage A (data not shown) and in five of the assemblage B-infected family clusters (Table 4). All members in each of these family clusters shared the same MLGs.

Treatment

Information on treatment of giardiasis was available for 173 patients: 108 had been given metronidazole, 63 tinidazole, and two albendazole. Due to the study design, consistent follow-up of treatment was not done. Nevertheless, 11 patients (seven given metronidazole and four tinidazole) provided a second fecal sample containing Giardia parasites after treatment. All three genes were investigated in samples taken before and after treatment, which revealed two patients with assemblage A, (MLG AII-1 or MLG AII-2 respectively), eight with assemblage B, and one with a mixed assemblage A+B infection. The sample provided after treatment by the patient with mixed infection was positive only for assemblage B. Both patients with assemblage A had been infected in Sweden, whereas the patient with mixed assemblage A+B and six of the patients with assemblage B had been infected in India. The remaining assemblage B patients were infected in Ecuador and Gambia, respectively.

Discussion

To the best of our knowledge, this is the first large-scale study of Giardia infections to use both sequence-based multilocus genotyping and assemblage-specific PCR to determine associations between symptoms and assemblages and to investigate transmission dynamics and possible zoonotic transmission. Most of the Giardia infections we studied (69%) were acquired abroad, although domestic infections were not infrequent (31%). Assemblage B was found in 128 patients (60%) and assemblage A in 73 (34%). Mixed assemblage A+B infections were identified in six patients (3%) while samples from seven patients (3%) could not be amplified in PCR. Assemblage B dominated in infections acquired in Sweden (64%) as well as in those originating from most other parts of the world, which agrees with previous reports from Europe, Latin America, Asia, and Australia [13,22,34-36], although studies from Brazil and Egypt have instead shown a predominance of assemblage A [37,38]. It has been suggested that the higher parasite excretion in assemblage B-infections as detected by microscopy [39] or real-time PCR [13] could explain the predominance of this assemblage in certain areas. Clearly, further research is needed to explore this issue.

Several studies have reported correlations between assemblages and symptoms, but there has been a lack of concordance in the data

Table 5. Probable area of origin of infection in 214 giardiasis patients presented in relation to assemblages.

	Sweden	Other European countries	Africa	Asiaª	Latin America	North America	Not stated	Total
Assemblage A	22	5	11	22	12	0	1	73
Assemblage B	42	11	20	43	9	3	0	128
Assemblage A+B	0	0	1	5		0	0	6
PCR negative	2	0	3	1	1	0	0	7
All cases	66	16	35	71	22	3	1	214

^aThirty-eight infections originating from Asia were acquired in India, and six of those were assemblage A, 28 assemblage B, and four assemblage A+B. doi:10.1371/journal.pntd.0001262.t005 **Table 6.** Symptoms of 145 giardiasis patients in relation to assemblage (mixed infections with other enteropathogens excluded).*

	Patients infected with Giardia intestinalis							
Symptoms	Assemblage A (n = 51)	Assemblage B (n = 87)	All patients (n = 145 ^c)					
Diarrhea ^a	48/51 (94)	86/87 (99)	141/145 (97)					
Bowel movements ^a								
<3/day	9/51 (18)	20/87 (23)	30/145 (21)					
3–5/day	12/51 (24)	31/87 (36)	44/145 (30)					
>5/day	27/51 (53)	35/87 (40)	67/145 (46)					
Abdominal pain ^a	33/50 (66)	55/85 (65)	94/142 (66)					
Bloody stools ^a	3/50 (6)	5/87 (6)	8/144 (6)					
Vomiting ^a	17/50 (34)	29/87 (33)	47/144 (33)					
Flatulence ^b	33/51 (65)	73/86 (85)	112/144 (78)					
Fever $>$ 38°C ^a	12/50 (24)	19/85 (22)	32/142 (23)					
Loss of weight ^a	33/50 (66)	65/85 (76)	103/142 (73)					

NOTE: Response rates for the different items on the questionnaires varied from 98% to 100%.

*The data given represent number of findings/number of patients who answered the specific questions (%).

^aNo significant difference between the assemblage A and B patient groups.

^bSignificant difference between the assemblage A and assemblage B patient

groups, p=0.006.

^cIncludes four patients with assemblage A+B and three with negative PCR results. doi:10.1371/journal.pntd.0001262.t006

obtained [12–16,40]. We compared symptoms and assemblages in 138 patients with pure assemblage A or B infections, and the only apparent correlation was that flatulence was significantly more common in children aged 0–5 years who were infected with assemblage B parasites. *Blastocystis* was detected in samples from many (14%) of our patients, but the clinical significance of this organism is controversial, and the occurrence had no effect on the outcome of the analyses (data not shown). In 11 family clusters, the members shared the same *Giardia* MLGs, but, interestingly, only some of the individuals developed symptoms. This indicates that attention should be given to other factors than the genotypes of parasites such as the patients' immune status, parasite infective dose, age, nutritional status, underlying systemic diseases, and presence of co-infections.

Recent studies using either conventional PCR with assemblagespecific primers [18] or real time PCR [41] have shown a much larger degree of mixed-assemblage infections in humans than was previously reported, which further complicates the attribution of symptoms to infection with a specific assemblage. In our investigation, the combined results of RFLP, sequencing, and use of assemblage-specific primers yielded only six patients (3%) with mixed-assemblage infections, which contradicts the abovementioned findings but agrees with another recent study that also employed assemblage-specific tpi primers [34]. These six patients were all infected outside Europe (one in Africa and five in Asia), in countries where Giardia transmission is much higher, and they were all symptomatic. However, five of them were co-infected with bacteria or other parasites, which underlines the difficulty of correlating Giardia genotypes with symptoms in cases of infection originating from endemic areas (Table 2).

Since 1979, the WHO has regarded giardiasis as a zoonotic disease, and many studies have suggested zoonotic transmission of *Giardia*, although this has never been proven conclusively. It has been proposed that multilocus genotyping is a powerful tool for

investigating possible zoonotic transmission [11,17], and we used this strategy in the present study of human isolates and in a previous investigation of Swedish animal isolates [20]. In the current study, it seemed that most patients with assemblage A infection had acquired their parasites through the anthroponotic route, since they harbored sub-assemblage AII, which is rarely found in animals [11]. However, three patients who were diagnosed with symptomatic giardiasis in November 2008 and came from the same area north of Stockholm carried two different assemblage A MLGs previously detected in cats and ruminants in Sweden (Table 3, Fig. 2A) [20]. Two of these patients did not report any particular contact with animals prior to infection, and, notably, they were the only patients infected with assemblage A that were hospitalized and treated with intravenous fluids during the study period. The third patient was apparently a moose and deer hunter who was responsible for dismembering the animal carcasses and whose hunting expedition took place shortly before he became symptomatic. None of these patients had traveled outside their home area prior to infection, thus domestic zoonotic transmission is strongly indicated.

Few assemblage B isolates have been found in animals in Sweden, and not many investigators have conducted multilocus genotyping of assemblage B in isolates from animals [19,42], hence it is difficult to compare human and animal isolates in this context. Two studies of the β-giardin locus in animal isolates have detected assemblage B subtypes identical to those that were most common in our study (i.e., B1-1 and B1-3) [43,44]. However, findings concerning only one gene are not sufficient to compare different isolates, which stresses the need for multilocus genotyping of more B isolates, from both humans and animals.

Eighteen family clusters were identified during the study period. Six of these were connected with assemblage A, and the families were infected with MLG AII-1 or MLG AII-2, or a combination of both. It should be mentioned that the resolution of assemblage A was limited, since these two MLGs predominated throughout the study (Table 3). Markers with higher discriminatory power are needed to render multilocus genotyping useful in outbreaks involving *Giardia* assemblage A. In five out of eight clusters where assemblage B parasites were identified, patients connected with each other carried the same assemblage B MLGs (Table 4), which indicates that in some instances it might be possible to use assemblage B MLGs for source tracing in outbreak situations. However, the frequent occurrence of overlapping nucleotides in assemblage B sequences in one, two, or all three genes examined limited the value of MLGs from assemblage B, because these could be determined in only 31 isolates exhibiting sequences without heterogeneous positions in all three genes (Table 4). In these isolates, there was good concordance between sub-assemblages BIII and BIV at the tpi and gdh loci, but otherwise the separation of isolates into sub-assemblages BIII or BIV was greatly affected by overlap occurring at the specific nucleotide positions that can supposedly distinguish between these subgroups [33] (Supplementary Tables S2A-C). This implies that determining BIII and BIV is of limited value due to widespread occurrence of heterogeneous positions in the sequences. The polymorphism of assemblage B was also reflected in the RFLPs of this assemblage, which sometimes were difficult to interpret due to unusual patterns caused by the presence of alternate nucleotides at the restriction sites (Fig. 1). A high degree of polymorphism in assemblage B has also been observed in other studies [17,22,26,45] and has been further investigated by cloning [46-48]. This feature has been attributed to mixed subtype infections or allelic sequence divergence, or a combination of both. In the present study, isolates for which all three genes produced double peaks in the

chromatogram were strongly correlated with infections acquired outside Europe (p = <0.001), suggesting that mixed subtype infections are more common in areas with high prevalence of *Giardia*. Almost all assemblage A sequences, except five with mixed subtype A2 and A3 at the β -giardin locus, displayed unequivocal nucleotide sequences at all three loci. The level of allelic sequence divergence was found to be higher in a recent whole-genome sequencing of the assemblage B isolate GS [49] compared to that observed in assemblage A isolate WB [50], which agrees with the larger number of double peaks for assemblage B sequences.

There was good agreement between assignment of assemblages at all three loci, and no assemblage swapping (i.e., different assemblages at different loci in the same isolate) was detected in any of the isolates exhibiting single-assemblage infections at all three genes, as determined by PCR-RFLP and sequencing. Assemblage swapping has been reported by other investigators [9,24,36] and has been attributed to recombination between assemblages or mixed assemblage infection. We identified six mixed assemblage infections and these can potentially be recombinants or fusions between assemblage A and B parasites but a serial dilution of the templates removed the minor assemblage, arguing against recombinat isolates (data not shown). Intra-assemblage recombination is more likely to occur than interassemblage recombination, since assemblage A and B isolates are only 78% identical at the nucleotide level. However, intraassemblage recombination is more difficult to detect but our data in Table 4 shows a mixed pattern of alleles, which can be the result of recombination between different assemblage B subtypes. Preliminary results from our group have revealed that overlapping nucleotides occur in discriminatory positions in the chromatograms of single Giardia cysts or trophozoites obtained from assemblage B isolates (data not shown). This demonstrates that there are different subtype-specific alleles in one cell, which further strengthens the suggestion that recombination occurs within assemblages.

Treatment failure was suspected in 11 patients (two with assemblage A, eight with assemblage B, and one with mixed assemblage infection) who showed persistence of parasites in a second fecal sample provided after treatment. Comparison of sequence data obtained before and after treatment was possible in 10 of those subjects. Notably, the patient with mixed assemblage A+B infection harbored only assemblage B parasites after treatment. The isolates from the eight patients with assemblage B infection had almost identical sequences before and after treatment, although the results varied slightly due to frequent double peaks in the chromatograms. None of these patients shared the exact same sequences at all loci, thus their Giardia isolates could not be attributed to any specific combination of assemblage B sequences. The present study was not designed to conduct systematic post-treatment follow-ups, and thus the 11 patients with suspected treatment failure were detected incidentally, and no additional data could be collected. Anecdotal reports concerning treatment failure in patients with assemblage B acquired in India have indicated that well-planned studies are needed to address this issue.

We performed multilocus genotyping of the three proteincoding genes ß-giardin, *tpi*, and *gdh* combined with analysis using assemblage-specific *tpi* primers, which at present might be the best choice for *Giardia* genotyping. However, it was difficult to compare our results with the findings of other investigations due to the lack of standardization of PCR primers and protocols, the extensive polymorphisms in assemblage B in the studied genes combined with inconsistent interpretation of sequence chromatograms, and no uniform nomenclature of subtypes and MLGs. Hopefully, the addition of new markers and thorough testing of a large number of reference strains and isolates from various sources using of the same loci and the same primer sets will aid the development of a more uniform, standardized, and informative approach to *Giardia* genotyping.

Most of the Giardia infections investigated here originated abroad, although the proportion of domestic infections was by no means insignificant. Even though most of our patients appeared to be infected by anthroponotic transmission, we found evidence that also zoonotic transmission of Giardia occurs in Sweden, since the same assemblage A MLGs previously detected in Swedish cats and ruminants were found in a few human domestic cases. The only correlation observed between symptoms and assemblages was that flatulence was more common in children 0-5 years of age infected with assemblage B. Mixed-assemblage infections were uncommon. Assemblage B was found more often than assemblage A in both travelers and domestic cases, and it was also more common in patients with suspected treatment failure. Determination of assemblage A and B MLGs proved to be a useful tool that can be used in outbreak situations or to demonstrate possible zoonotic potential. However, the extensive polymorphism seen in assemblage B hampered the determination of MLGs, because only isolates without ambiguous positions could be included.

Supporting Information

Figure S1 Nucleotide maximum likelihood trees based on ßgiardin (A), *gdh* (B), and *tpi* (C) gene sequences. All unique unambiguous sequences from Table 4 are included together with reference isolates and isolates from our previous MLG study of animals in Sweden [20] (Supplementary Table S1). The color coding is the same as for the clustering into BIII (red) and BIV (blue) isolates shown in Figure 2. The trees are based on 475, 393, and 490 aligned positions, respectively. Only bootstrap support values >50 are shown.

(EPS)

File S1 β -giardin sequences from 120 isolates. (DOC)

File S2 Gdh sequences from 120 isolates. (DOC)

File S3 Tpi sequences from 120 isolates.

(DOC)

Table S1

(DOC)

Table S2

(DOC)

Acknowledgments

Ulf Emanuelsson, SLU, is acknowledged for help with statistical analyses.

Author Contributions

Conceived and designed the experiments: ML JOA BS SB-K SGS. Performed the experiments: ML IP LK JOA. Analyzed the data: ML IP SB-K JOA BS SGS. Contributed reagents/materials/analysis tools: ML LK IP JOA BS. Wrote the paper: ML JOA BS SGS.

References

- Lasek-Nesselquist E, Welch DM, Sogin ML (2010) The identification of a new Giardia duodenalis assemblage in marine vertebrates and a preliminary analysis of G. duodenalis population biology in marine systems. Int J Parasitol 40: 1063–1074.
- Thompson RC (2000) Giardiasis as a re-emerging infectious disease and its zoonotic potential. Int J Parasitol 30: 1259–1267.
- WHO (1996) The World Health Report 1996. Available from: http://www. who.int/whr/1996/en/index.html.
- Savioli L, Smith H, Thompson A (2006) Giardia and Cryptosporidium join the 'Neglected Diseases Initiative'. Trends Parasitol 22: 203–208.
- Monis PT, Andrews RH, Mayrhofer G, Ey PL (2003) Genetic diversity within the morphological species Giardia intestinalis and its relationship to host origin. Infect Genet Evol 3: 29–38.
- WHO (1979) Parasitic zoonoses. Report of a WHO expert committee with the participation of FAO. Available from: http://whqlibdoc.who.int/trs/ WHO_TRS_637.pdf.
- Inpankaew T, Traub R, Thompson RC, Sukthana Y (2007) Canine parasitic zoonoses in Bangkok temples. Southeast Asian J Trop Med Public Health 38: 247–255.
- Traub RJ, Monis PT, Robertson I, Irwin P, Mencke N, et al. (2004) Epidemiological and molecular evidence supports the zoonotic transmission of Giardia among humans and dogs living in the same community. Parasitology 128: 253–62.
- Caccio SM, Ryan U (2008) Molecular epidemiology of giardiasis. Mol Biochem Parasitol 160: 75–80.
- Monis PT, Thompson RC (2003) Cryptosporidium and Giardia-zoonoses: fact or fiction? Infect Genet Evol 3: 233–244.
- Sprong H, Caccio SM, van der Giessen JW (2009) Identification of zoonotic genotypes of Giardia duodenalis. PLoS Negl Trop Dis 3: e558.
- Aydin AF, Besirbellioglu BA, Avci IY, Tanyuksel M, Araz E, et al. (2004) Classification of Giardia duodenalis parasites in Turkey into groups A and B using restriction fragment length polymorphism. Diagn Microbiol Infect Dis 50: 147–151.
- Haque R, Roy S, Kabir M, Stroup SE, Mondal D, et al. (2005) Giardia assemblage A infection and diarrhea in Bangladesh. J Infect Dis 192: 2171–2173.
- Homan WL, Mank TG (2001) Human giardiasis: genotype linked differences in clinical symptomatology. Int J Parasitol 31: 822–826.
- Read C, Walters J, Robertson ID, Thompson RC (2002) Correlation between genotype of Giardia duodenalis and diarrhoea. Int J Parasitol 32: 229–231.
- Sahagun J, Clavel A, Goni P, Seral C, Llorente MT, et al. (2008) Correlation between the presence of symptoms and the Giardia duodenalis genotype. Eur J Clin Microbiol Infect Dis 27: 81–83.
- Caccio SM, Beck R, Lalle M, Marinculic A, Pozio E (2008) Multilocus genotyping of Giardia duodenalis reveals striking differences between assemblages A and B. Int J Parasitol 32: 1023–1531.
- Geurden T, Levecke B, Caccio SM, Visser A, De Groote G, et al. (2009) Multilocus genotyping of Cryptosporidium and Giardia in non-outbreak related cases of diarrhoea in human patients in Belgium. Parasitology 136: 1161–1168.
- Levecke B, Geldhof P, Claerebout E, Dorny P, Vercammen F, et al. (2009) Molecular characterisation of Giardia duodenalis in captive non-human primates reveals mixed assemblage A and B infections and novel polymorphisms. Int J Parasitol 39: 1595–1601.
- Lebbad M, Mattsson JG, Christensson B, Ljungstrom B, Backhans A, et al. (2010) From mouse to moose: multilocus genotyping of Giardia isolates from various animal species. Vet Parasitol 168: 231–239.
- Svenungsson B, Velicko I, Petersson I, De Jong B, Andersson Y, et al. (2007) [Giardiasis as differential diagnosis in diarrhea outbreaks in child day centers. Written hygienic guidelines and adequate testing can reduce the transmission]. Lakartidningen 104: 500–503.
- Lebbad M, Ankarklev J, Tellez A, Leiva B, Andersson JO, et al. (2008) Dominance of Giardia assemblage B in Leon, Nicaragua. Acta Trop 106: 44–53.
- Lalle M, Pozio E, Capelli G, Bruschi F, Crotti D, et al. (2005) Genetic heterogeneity at the beta-giardin locus among human and animal isolates of Giardia duodenalis and identification of potentially zoonotic subgenotypes. Int J Parasitol 35: 207–213.
- Read CM, Monis PT, Thompson RC (2004) Discrimination of all genotypes of Giardia duodenalis at the glutamate dehydrogenase locus using PCR-RFLP. Infect Genet Evol 4: 125–130.
- Sulaiman IM, Fayer R, Bern C, Gilman RH, Trout JM, et al. (2003) Triosephosphate isomerase gene characterization and potential zoonotic transmission of Giardia duodenalis. Emerg Infect Dis 9: 1444–1452.

- Lalle M, Bruschi F, Castagna B, Campa M, Pozio E, et al. (2009) High genetic polymorphism among Giardia duodenalis isolates from Sahrawi children. Trans R Soc Trop Med Hyg 103: 834–838.
- Geurden T, Geldhof P, Levecke B, Martens C, Berkvens D, et al. (2008) Mixed Giardia duodenalis assemblage A and E infections in calves. Int J Parasitol 38: 259–264.
- 28. Caccio SM, De Giacomo M, Pozio E (2002) Sequence analysis of the betagiardin gene and development of a polymerase chain reaction-restriction fragment length polymorphism assay to genotype Giardia duodenalis cysts from human faecal samples. Int J Parasitol 32: 1023–1030.
- Stamatakis A, Ludwig T, Meier H (2005) RAxML-III: a fast program for maximum likelihood-based inference of large phylogenetic trees. Bioinformatics 21: 456–463.
- Abe N, Tanoue T, Noguchi E, Ohta G, Sakai H (2010) Molecular characterization of Giardia duodenalis isolates from domestic ferrets. Parasitol Res 106: 733–736.
- Robertson LJ, Forberg T, Hermansen L, Hamnes IS, Gjerde B (2007) Giardia duodenalis cysts isolated from wild moose and reindeer in Norway: genetic characterization by PCR-rflp and sequence analysis at two genes. J Wildl Dis 43: 576–585.
- Trout JM, Santin M, Greiner E, Fayer R (2004) Prevalence of Giardia duodenalis genotypes in pre-weaned dairy calves. Vet Parasitol 124: 179–186.
- Wielinga CM, Thompson RC (2007) Comparative evaluation of Giardia duodenalis sequence data. Parasitology 134: 1795–1821.
- Breathnach AS, McHugh TD, Butcher PD (2010) Prevalence and clinical correlations of genetic subtypes of Giardia lamblia in an urban setting. Epidemiol Infect. pp 1–9.
- van der Giessen JW, de Vries A, Roos M, Wielinga P, Kortbeek LM, et al. (2006) Genotyping of Giardia in Dutch patients and animals: a phylogenetic analysis of human and animal isolates. Int J Parasitol 36: 849–858.
- Yang R, Lee J, Ng J, Ryan U (2010) High prevalence Giardia duodenalis assemblage B and potentially zoonotic subtypes in sporadic human cases in Western Australia. Int J Parasitol 40: 293–297.
- Abdel-Moneim SM, Sultan DM (2008) Genetic characterization of Giardia lamblia isolates from Egyptian patients with relation to clinical giardiasis. J Egypt Soc Parasitol 38: 547–560.
- Volotao AC, Costa-Macedo LM, Haddad FS, Brandao A, Peralta JM, et al. (2007) Genotyping of Giardia duodenalis from human and animal samples from Brazil using beta-giardin gene: a phylogenetic analysis. Acta Trop 102: 10–19.
- Kohli A, Bushen OY, Pinkerton RC, Houpt E, Newman RD, et al. (2008) Giardia duodenalis assemblage, clinical presentation and markers of intestinal inflammation in Brazilian children. Trans R Soc Trop Med Hyg 102: 718–725.
- Gelanew T, Lalle M, Hailu A, Pozio E, Caccio SM (2007) Molecular characterization of human isolates of Giardia duodenalis from Ethiopia. Acta Trop 102: 92–99.
- Almeida A, Pozio E, Caccio SM (2010) Genotyping of Giardia duodenalis cysts by new real-time PCR assays for detection of mixed infections in human samples. Appl Environ Microbiol 76: 1895–1901.
- Beck R, Sprong H, Bata I, Lucinger S, Pozio E, et al. (2011) Prevalence and molecular typing of Giardia spp. in captive mammals at the zoo of Zagreb, Croatia. Vet Parasitol 175: 40–46.
- Solarczyk P (2009) [Occurrence of Giardia species and genotypes in humans and animals in Wielkopolska region, Poland]. Wiad Parazytol 55: 459–462.
- Winkworth CL, Learmonth JJ, Matthaei CD, Townsend CR (2008) Molecular characterization of Giardia isolates from calves and humans in a region in which dairy farming has recently intensified. Appl Environ Microbiol 74: 5100–5105.
- Robertson LJ, Forberg T, Hermansen L, Gjerde BK, Langeland N (2007) Molecular characterisation of Giardia isolates from clinical infections following a waterborne outbreak. J Infect 55: 79–88.
- Hussein AI, Yamaguchi T, Nakamoto K, Iseki M, Tokoro M (2009) Multiplesubgenotype infections of Giardia intestinalis detected in Palestinian clinical cases using a subcloning approach. Parasitol Int 58: 258–262.
- Kosuwin R, Putaporntip C, Pattanawong U, Jongwutiwes S (2010) Clonal diversity in Giardia duodenalis isolates from Thailand: evidences for intragenic recombination and purifying selection at the beta giardin locus. Gene 449: 1–8.
- Lasek-Nesselquist E, Bogomolni AL, Gast RJ, Welch DM, Ellis JC, et al. (2008) Molecular characterization of Giardia intestinalis haplotypes in marine animals: variation and zoonotic potential. Dis Aquat Organ 81: 39–51.
- 49. Franzen O, Jerlstrom-Hultqvist J, Castro E, Sherwood E, Ankarklev J, et al. (2009) Draft genome sequencing of giardia intestinalis assemblage B isolate GS: is human giardiasis caused by two different species? PLoS Pathog 5: e1000560.
- Morrison HG, McArthur AG, Gillin FD, Aley SB, Adam RD, et al. (2007) Genomic minimalism in the early diverging intestinal parasite Giardia lamblia. Science 317: 1921–1926.